# Care home residents' quality of life and its association with CQC ratings and workforce issues: the MiCareHQ mixed-methods study

Ann-Marie Towers,<sup>1\*</sup> Nick Smith,<sup>2</sup> Stephen Allan,<sup>2</sup> Florin Vadean,<sup>2</sup> Grace Collins,<sup>2</sup> Stacey Rand,<sup>2</sup> Jennifer Bostock,<sup>3</sup> Helen Ramsbottom,<sup>3</sup> Julien Forder,<sup>2</sup> Stefania Lanza<sup>4</sup> and Jackie Cassell<sup>4</sup>

\*Corresponding author A.Towers@kent.ac.uk

**Declared competing interests of authors:** Ann-Marie Towers is a member of the National Institute for Health Research (NIHR) Research for Patient Benefit Funding Committee (2018 to present) and a member of the NIHR Research for Social Care Funding Committee (2018 to present). Jennifer Bostock is a member of the NIHR Health Services and Delivery Research Funding Committee. Helen Ramsbottom is a member of the NIHR Health Technology Assessment Prioritisation Committee.

**Disclaimer:** This report contains transcripts of interviews conducted in the course of the research, or similar, and contains language which may offend some readers.

Published October 2021 DOI: 10.3310/hsdr09190

## Scientific summary

The MiCareHQ mixed-methods study
Health Services and Delivery Research 2021; Vol. 9: No. 19
DOI: 10.3310/hsdr09190

NIHR Journals Library www.journalslibrary.nihr.ac.uk

<sup>&</sup>lt;sup>1</sup>Centre for Health Services Studies, University of Kent, Canterbury, UK <sup>2</sup>Personal Social Services Research Unit, University of Kent, Canterbury, UK <sup>3</sup>Public and patient involvement representative, Canterbury, UK <sup>4</sup>Brighton and Sussex Medical School, Brighton, UK

## **Scientific summary**

#### **Background**

Over 425,000 older people in England live in care homes because they have significant long-term health problems. Many are living with dementia. The Care Act 2014 (Great Britain. *The Care Act 2014*. London: The Stationery Office; 2014) emphasises the importance of measuring and improving the well-being of social care users. However, there is no single, agreed minimum data set in the UK, making it difficult to measure and monitor the health and social care outcomes of residents as a quality indicator.

To ensure the quality of care provided and to inform public choice, care homes are regulated by the Care Quality Commission (CQC). Inspection reports and quality ratings are publicly available, with homes awarded a quality rating ('outstanding', 'good', 'requires improvement' or 'inadequate'). The most recent system rated around one-fifth of homes as 'inadequate' or 'requires improvement'. There is very little evidence concerning how well these ratings are associated with residents' quality of life and the characteristics and employment conditions of the workforce.

Care home quality varies for many reasons, but the nature and characteristics of the workforce, and the approaches of that workforce to care, are likely to be major determinants. Although a small body of literature exists on the impact of workforce characteristics (e.g. staff turnover) on care home quality, most of this literature is US based and focuses on clinical markers of quality or other process measures and not on residents' quality of life. Furthermore, there is very little statistical evidence in England linking skill mix and training levels with care quality outcomes.

Rigorous and reliable methods are needed to understand how well residents' needs are being met. The Adult Social Care Outcomes Toolkit (ASCOT) is a suite of tools for measuring the aspects of quality of life that may be supported by social care. We call this 'social care-related quality of life' (SCRQoL). ASCOT offers different modes of completion, including a mixed-methods approach to measuring SCRQoL in care homes [Adult Social Care Outcomes Toolkit Care Homes, four levels (ASCOT-CH4)]. This was designed to be used with residents who cannot self-report. Trained researchers collect evidence about SCRQoL through observations, conversations with residents and interviews with staff and family members.

ASCOT was designed to measure residents' current SCRQoL and also the impact of services (e.g. care homes) on residents' SCRQoL. However, it does not measure aspects of health, such as pain, anxiety or depression, that are reported to be under-recognised and undertreated in care homes.

The Measuring and Improving Care Home Quality (MiCareHQ) study focused on care home quality. It brought together three interlinked strands of work to examine the relationship between care home residents' health and quality of life, regulator quality ratings and the skill mix and employment conditions of the workforce.

#### **Objectives**

Specifically, the study aimed to:

- 1. develop and test measures of pain, anxiety and depression for residents unable to self-report
- 2. assess the extent to which the CQC's quality ratings are consistent with indicators of residents' quality of life
- 3. assess the relationship between aspects of the staffing of care homes and the quality of care homes.

#### Methods and results

The aims of the study were addressed through a mixed-methods design involving three interlinked work packages.

Ethics approval was required for the primary data collection undertaken in work packages 1 and 2. The study received ethics approval from the Health Research Authority, which governs research ethics in the UK (reference 18/LO/0657). As this was a social care study, approval was sought from and granted by the Association of Directors of Adult Social Services and each participating local authority.

#### Public involvement and engagement strategy

Public involvement and engagement was an integral part of the study and was delivered in the following ways:

- 1. Study Steering Group two members of the public with experience of social care were recruited to the Study Steering Committee as lay representatives.
- 2. Co-researchers in work package 1 we recruited three lay co-researchers to assist with the focus groups with staff and to contribute to the development of the new measures of pain, anxiety and depression. This is described in detail in *Chapter 3* of the main report.
- 3. Contributions to study outputs lay advisors from the Study Steering Group contributed to the drafting of the *Plain English summary* of this report and reviewed it for quality and clarity of writing. The co-researchers from work package 1 co-authored *Chapter 7* of the main report.

#### Work package 1: measures

#### Methods

A rapid review of existing measures, with a particular focus on tools already incorporating observational methods, was undertaken to inform the development of items of pain, anxiety and depression to be used alongside the ASCOT-CH4. Three focus groups with care home staff were held to (1) identify how staff recognise when residents are in pain or are feeling anxious or depressed; (2) identify the words staff use to describe residents' anxiety, pain and depression; and (3) review draft questions on pain, anxiety and depression.

Following the focus groups, the measures were revised and cognitively tested with a sample of staff and family members of care home residents (n = 37) using a combination of verbal probing techniques and thinking aloud. Lay co-researchers were involved in the focus groups and subsequent revisions of the new measures.

#### **Results**

The rapid review, reported in *Chapter 2* of the main report, yielded 196 unique papers on pain tools and 166 unique papers on anxiety and depression tools after duplicates were removed. The titles and abstracts of all of these papers were then reviewed for relevance and adherence to inclusion criteria. In the final review, 26 papers were included for pain measures and 22 papers were included for measures of anxiety and depression, yielding 22 tools for each. The review revealed that most tools relied on a maximum of two modes of data collection (e.g. self-report, proxy or observation). Tools developed for people living with more severe dementia often rely on proxy and observation, thereby excluding the person's own voice. Very few anxiety and depression tools had been developed for people living with more severe dementia. None of the existing tools estimated the impact that care homes have on residents' pain, anxiety and depression. Informed by the review, the research team drafted questions and response options for each domain (pain, anxiety and depression), following the approach used in ASCOT.

Twenty-two care home staff took part in the three focus groups (see *Chapter 3* of the main report). Most of these were care workers (n = 11) or care home nurses (n = 6); two were managers and three had other roles in the home (e.g. activity co-ordinators). Staff were able to describe the verbal and non-verbal (behavioural) signs of pain, anxiety and depression among residents, which helped us conceptualise the observable indicators relevant to our tools. Staff felt that residents rarely used terms that had clinical connotations, such as 'depressed', and preferred more colloquial terms, such as 'low mood' or 'feeling sad or down'.

We conducted three iterative rounds of cognitive interviews with 21 care home staff and 16 family members. Cognitive testing led to two important changes: (1) the outcome states were no longer anchored to 'never' because staff and family members felt that everyone feels anxious or down or experiences pain some of the time; and (2) the depression domain was renamed as low mood, reflecting the emphasis on quality of life rather than on clinical or diagnostic criteria. The question wording, response options and observational guidance were finalised for the three new items of pain, anxiety and low mood, which were to be piloted alongside the ASCOT in work package 2.

#### Work package 2: psychometric testing

#### Methods

We undertook primary data collection to pilot and psychometrically test the new items and to explore the feasibility of the mixed-methods approach (see *Chapter 4* in the main report), recruiting 182 residents from 20 care homes for older adults (10 nursing and 10 residential) from four local authorities in South East England. To explore the construct validity of the new domains of pain, anxiety and low mood, questionnaires also contained staff-rated validated scales and items relating to these concepts so that we could explore the hypothetical relationships of these with the new attributes in the analysis.

#### **Results**

An examination of missing data revealed that the mixed-methods approach was necessary, and there was also tentative evidence of its feasibility. As expected, numbers of missing data were large for resident self-report interviews, with fewer than one-quarter of residents responding to any single item of the structured interview. However, data were still collected through qualitative interviews, observations and proxy interviews, which informed final researcher ratings, and so every resident had a rating for all items.

There was good evidence of the construct validity of the three new items for both 'current' and 'expected' ratings, meaning that they appeared to be capturing the constructs we intended. Psychometric testing revealed that, although the eight items of the ASCOT-CH4 formed a single unidimensional measurement scale, the three new items did not. This was expected, as the concepts captured by the new items (i.e. pain, low mood and anxiety) relate to aspects of health-related quality of life, which is distinct from the ASCOT construct of SCRQoL. Analysis also revealed that pain did not fit well into a measurement scale alongside low mood and anxiety. Taken together, these results indicate that the three items are better conceptualised as separate 'modules' that relate to the concepts of psychological health and pain, which may be added flexibly alongside ASCOT-CH4 (as a separate scale), with low mood and anxiety combined and pain standing alone.

#### Work package 3: care home quality

#### Methods

First, we combined data from two studies to examine the relationship between CQC quality ratings and residents' SCRQoL, controlling for confounding variables. The first study used data collected in work package 2 of MiCareHQ (see *Chapter 4* of the main report) and the second was a previous study, Measuring Outcomes Of Care Homes (MOOCH; see *Chapter 5* of the main report). Both studies had a cross-sectional design, in which researchers spent time in each care home and used the ASCOT-CH4 to

carry out observations of and interviews with staff and (where possible) residents and family members. Additional data about residents were collected from staff using questionnaires, including demographic information, health status and ability to complete activities of daily living.

Next, we conducted an analysis of secondary data of English care homes from 2016 to 2018 to model the relationship between CQC quality ratings and workforce characteristics. We focused on the effect that training provision for staff and staff terms and conditions (wages and turnover/vacancy rates) had on the quality of care homes. All of these variables, which can be affected by policy, are expected to be important determinants of quality. We used the following econometric methods to assess the relationship between quality and workforce characteristics: longitudinal panel data models, multiple imputation to address missing data, and an instrumental variable approach to control for potential endogeneity between quality and workforce characteristics, in particular staff wages.

#### Results

As described in *Chapter 5* of the main report, we found a significant positive association between residents' SCRQoL and CQC quality ratings. Everything else being equal, homes rated 'good' or 'outstanding' were associated with a 12% improvement in mean current SCRQoL among residents who had higher levels of dependency. Although care homes generally met the residents' needs in basic aspects of care, 'good' and 'outstanding' homes made a meaningful difference to those aspects of residents' lives that add quality to their days: feeling in control, being engaged in activities, being socially fulfilled and being treated with dignity.

Using the national sample of care homes (n = 12,052 observations of 5555 care homes), we found a significant positive association between wages and the prevalence of training in both dementia and dignity-/person-centred care and the likelihood of higher quality, whereas high staff turnover and job vacancy rates had a significant negative association. This was particularly true for wages, as we found that a 10% increase in the average care worker wage increased the likelihood of a 'good' or 'outstanding' rating by 7%. This is reported in *Chapter 6* of the main report.

These findings are of significant interest to the public, prospective residents, family members, care providers and the care regulator, the CQC.

#### **Conclusions**

Improving working conditions (e.g. wages and training) and reducing staff turnover is one way to increase care quality and outcomes for residents. This is particularly important for care home residents with the greatest needs, who gain the most from homes rated 'good' or 'outstanding', especially with respect to social participation and feeling in control of their daily lives. However, measuring their needs and outcomes is challenging, as many cannot self-report. We have presented a robust method of measuring the SCRQoL of care home residents, using the ASCOT-CH4. We found this methodology to be feasible and the eight ASCOT items formed a unidimensional scale of SCRQoL. We applied this methodology to new domains of pain, anxiety and low mood and found that the mixed-methods approach could be applied to these domains, while still capturing the underlying constructs. The three additional domains could be used alongside the ASCOT in future to capture the impact of care homes on these important health-related aspects of quality of life.

This study has demonstrated the need for the systematic and consistent collection of data on residents' outcomes (e.g. SCRQoL) as an important indicator of care quality. This important evidence gap is being addressed in the ongoing Developing resources And minimum data set for Care Homes' Adoption (DACHA) study (http://dachastudy.com/; NIHR127234), which aims to develop and pilot a prototype minimum data set of care homes in England.

#### Implications for social care

- Potential mechanisms for improving residents' outcomes are strong leadership, a focus on continued
  quality improvement and a culture of care that gives staff time to listen to residents and meet their
  needs with compassion.
- This study found a strong and positive relationship between staff training and wages and care quality ratings.
- There is a need for the systematic and consistent collection of data on care home resident outcomes to allow a robust quantitative analysis.
- It is important to use innovative methods to capture the views and experiences of those with the greatest needs.

#### **Recommendations for research**

Future research should:

- 1. consider using a mixed-methods approach to data collection in care homes that includes the voices and experiences of those unable to self-report
- 2. explore the relationship between pain, anxiety and low mood and other indicators of care home quality
- 3. explore the impact of bottom-rated ('inadequate') and top-rated ('outstanding') homes on residents' outcomes
- 4. assess the productivity of staff by examining the relationship between resident outcomes and staffing characteristics and employment conditions (e.g. wages and training) directly.

#### **Funding**

This project was funded by the National Institute for Health Research (NIHR) Health Services and Delivery Research programme and will be published in full in *Health Services and Delivery Research*; Vol. 9, No. 19. See the NIHR Journals Library website for further project information.

### **Health Services and Delivery Research**

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

#### **HS&DR** programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-services-and-delivery-research.htm

#### This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 15/144/51. The contractual start date was in July 2017. The final report began editorial review in August 2020 and was accepted for publication in March 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

Copyright © 2021 Towers et al. This work was produced by Towers et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Scientific Adviser (Evidence Use), Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk